EFO Group Meeting	8/4/2019
Schmidt, Vanwingeren, Mazzio, Flood, Berkland, Wyncott, Moonier, Fieramosca,White, Rossi
Meetings this week: 
Shutdown Planning- Wednesday: Clay White.  Bruno will take phone during meeting
28-ID Utilities meeting-Shane Flood
2-ID Cant Meeting-Dean Wyncott
Mohan’s talk on the BM FE work for the upgrade will be on Thursday Aug 8th at 2pm in the 433 conference room.  Please plan to attend.
[bookmark: _GoBack]Betsy Dunn and Diane Wilkison will give a talk to us on CDB (component database) and e Traveler system on 8/12/19 at 2pm in 431.
ESH walkthroughs scheduled:  8/12/2019 11am for Sector 7
8/12/19 1pm for DND-CAT
Paul- First Aid case.  A DEEI had a sliver in his finger from opening an equipment box.  
New Head of ESH is David Gibson. 
Information on Incidents can be found here: http://inside.anl.gov/pages/reports-incident 
Liz- Soil Custodian-Rick Gammons came and Liz signed compliance agreement.  Once permit is in her name she will update web pages, signage and labels.
ESAFs: 32-ID New Laser Setup.  Elroy will take a look at it. Starts 8/8  Laser will not be used with the energetic material
Sector 7: Jet Fuel experiment.   Rossi and Fries will look at set up
Bruno-Mezzaine installation will start tomorrow.   Andy Stevens will do the WR.  No CCWP needed as no RSS involved.  Andy will also get blind penetration permit.  Truck lock is still full with items for the mezzanine install.

From Patti: 2-ID CANT Project to start in shutdown,  will let us know when RSS tags should be removed.  Look for WRs and a chance to use e traveller system.   
25-IDA start in November 2019
28-ID Utilities- Mead Electric won the contract.  Work to start in the near future. 
Wendy- 2 Rad experiments this week.  16BM and 16-ID.  
6-ID SV went well.  Door was repaired by Scheck and will be surveyed in the future.
Shane- 22-BM-D Strobe light wasn’t working.  Replaced with upgrade LED CCWP Closed out
9-ID Energetic Material.  Working to get samples shipped out
32-ID Energetic Material will be shipped in the next day or two 
SWITCHGEAR MAINTENANCE: X-6 shutdown on 9/6, CNM shutdown on 8/24
 AUTOMATIC TRANSFER SWITCH TESTING: 8/29 
APS Evacuation Drill: 8/27 at 9am



John- 23-BM is having BLEPS issues with their vacuum and as of right now can not take beam.  AR posted for delay in the scheduled Routine Rad survey.  Will be completed once BLEPS issue has been resolved.  
Working on 3rd party WPC/JSA for Sector 18.  Eppendorf to work on Centifuge
Lisa- Made first announcement in MCR.  Last official day is Aug 20 but will be on vacation that day.
Dean- 6-ID Shielding Verification went well.  Door was repaired by Scheck
Working on 2 third party WPC/JSA  Shamrock to install small workstation in 35-IDC Laser Control Room
and Contiunium for maintance and calibration of laser in 35-IDC and 35-IDB
   2-ID CANT meeting held weekly.  See Shane’s email for updates. Work will be done during shutdown. Last 2 Routine Rad surveys scheduled for this week
Gentle Reminder:  
Check your area ESAFs in the table vs the beamline cabinets.  Please be diligent about deactivating ESAFs that can be and removing them from the cabinet and table.
Also after any studies please make sure you reset PSS faults/trips that occurred from work in your area before the start of User Beam.   This will eliminate extra calls during a busy time.
Work Requests are coming in for shutdown work.  Please approve your areas as they come in.
Action Items:
Clay and John-continue to follow up on Paul Rossi's request for compressed gases in the 434/435 truck lot.  See if Oxidizer cabinet can be ordered and placed 20 feet away from other gasses.

